Author’s response to reviews

Title: Salaried and voluntary Community Health Workers: exploring how incentives and expectation gaps influence motivation

Authors:

Hermen Ormel (h.ormel@kit.nl)
Maryse Kok (Maryse.Kok@kit.nl)
Sumit Kane (sumitkane.gipe@gmail.com)
Rukhsana Ahmed (Rukhsana.Ahmed@lstmed.ac.uk)
Kingsley Chikaphupha (kingsley@reachtrust.org)
Sabina Faiz Rashid (sabina.frashid@gmail.com)
Daniel Gemechu (danieljohn42@yahoo.com)
Lilian Otiso (lilian.otiso@lvcthealth.org)
Mohsin Sidat (mmsidat@gmail.com)
Sally Theobald (Sally.Theobald@lstmed.ac.uk)
Miriam Taegtmeyer (Miriam.Taegtmeyer@lstmed.ac.uk)
Korrie de Koning (k.d.koning@kit.nl)

Version: 1 Date: 30 May 2019

Author’s response to reviews:

Dear James,

We read with interest the reviewers’ comments on our paper. We have addressed all of these in the table included in the separate cover document. We thank the reviewers for their insights and believe that the changes we have made have improved the paper. We hope it now in a form worthy of publication in HRH.

Yours sincerely,

Hermen Ormel (lead author)